Author's response to reviews

Title: Dental anxiety and dental attendance among 25-year-olds in Norway: time trends from 1997 to 2007

Authors:

Anne N Astrom Dr. (anne.nordrehaug@cih.uib.no)
Erik Skaret (erik.skaret@odont.uio.no)
Ola Haugejorden (ola.haugejorden@iko.uib.no)

Version: 2 Date: 25 February 2011

Author's response to reviews:

Bergen 26.02.11.

Dear Editor!

Thank's for the reviewers' report. We hereby submit a revised version of the ms "Dental anxiety and dental attendance among 25-year-olds in Norway: time trends from 1997 to 2007. The comments of the reviewers have been fylly taken into account in the revised ms.

1) The word subjects have been replaced with participants

2) The line 1 and 2 in the abstract has been removed

3) editorial changes have been made as suggested

4)Simple random sample has been explained (using individuals as primary sampling unit instead of for instance clusters)

Sincerely

Anne Nordrehaug Åstrøm